

# Species richness response to human pressure hides important assemblage transformations

Victor Cazalis

# ► To cite this version:

Victor Cazalis. Species richness response to human pressure hides important assemblage transformations. Proceedings of the National Academy of Sciences of the United States of America, 2022, 119 (19), pp.e2107361119. 10.1073/pnas.2107361119. hal-04889508

# HAL Id: hal-04889508 https://hal.science/hal-04889508v1

Submitted on 22 Jan 2025

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License

Title: Species richness response to human pressure hides important assemblage transformations

## Author: Victor Cazalis<sup>1,2,3</sup>\*

<sup>1</sup>German Centre for Integrative Biodiversity Research (iDiv) Halle-Jena-Leipzig, Puschstr. 4, D-04103 Leipzig, Germany

<sup>2</sup>Leipzig University, Ritterstraße 26, 04109 Leipzig, Germany

<sup>3</sup>CEFE, Univ Montpellier, CNRS, EPHE, IRD, Montpellier, France

\* Corresponding author. Email: victor.cazalis@idiv.de

#### Abstract:

Human activities' negative impact on biodiversity is undisputed, but debate remains vivid on their effect on species richness, a key index in ecology and conservation. While some studies suggest species richness declines with human pressure, others show that it can be insensitive or even respond positively to some human pressure, because some species ("losers") are replaced by others ("winners"). However, many "winners" are favoured by intermediate pressure but decline when pressure becomes too high, and we can thus expect species richness to decline above a certain human pressure. Analysing eBird data in tropical forests, I find that, under a certain threshold, increasing human footprint causes important composition changes with "losers" (habitat specialist, endemic, sensitive and threatened species) being replaced by "winners" (habitat non-specialist, large-range, human-tolerant, anthropophilic and non-native species), resulting in a slight increase in species richness. Above this threshold though, richness in "winners" drops (except for anthropophilic and non-native species), leading to a steep decline in overall species richness. I find that the shape of species richness response to human footprint varies between regions (comparing results from the North America Breeding Bird Survey, PREDICTS database, and eBird data across eight biodiversity hotspots) and identify five different trajectories in species richness response to human pressure. I suggest they can be classified depending on their slope and monotony in the "Replace then remove framework", unifying contradictory effects of human pressure on species richness.

#### Significance statement:

Human activities are causing biodiversity loss but there is still strong debate on their effect on species richness. Here I propose a unification of five trajectories of species richness response to increasing human pressure under the "Replace then remove framework". It consists in a first phase of assemblage transformation (with the replacement of "loser" by "winner" species), often followed by a second phase of steep decline in species richness (with the decline of many "winner" species) when human pressure exceeds a certain threshold. The empirical results presented in this study provide an outstanding illustration of assemblage transformations that may cause biotic homogenisation, demonstrating how habitat specialist, endemic, sensitive and threatened species are replaced by others with increasing human pressure.

# Introduction

The overall negative impact of human activities on wildlife species is widely recognised<sup>1</sup>. However, not all species are negatively impacted by human pressure<sup>2-4</sup>, and their response strongly depends on their traits and characteristics. For instance, many studies report that increasing human pressure causes the decline of most habitat specialist species, while many generalists can benefit from this pressure, resulting in biotic homogenisation<sup>5–7</sup>. Similarly, species that have a small geographic range tend to decrease with human pressure while species with a large range tend to respond more positively<sup>8</sup>, thus replacing small-range species across time<sup>9,10</sup>. As a result, some species are threatened with extinction because they are particularly impacted by such activities, while other species are currently expanding, sometimes helped by human introduction as non-native species<sup>1</sup>. Overall, this highlights that some species are negatively affected by human pressure (hereafter "losers"), while some can benefit from it ("winners").

Because of this heterogeneity in individual species response to human activities, there is uncertainty regarding the effect that human pressure has on the local number of species present in an assemblage – the species richness – which remains the most used biodiversity index in ecology and conservation. It is often expected that local extinctions of losers exceed colonisation by winners, and thus that species richness decreases with human pressure, as evidenced by several empirical studies<sup>11,12</sup>. Contrastingly, numerous studies found that, although assemblage composition is strongly affected by human pressure, species richness often remains stable through time<sup>13–16</sup>. These studies showed that winners are replacing losers as they go locally extinct, suggesting that local species richness may be insensitive to human pressure<sup>14,17</sup>. Some studies even reported that assemblage transformation leads to an increase in species richness through time<sup>9</sup>, or that species richness can be higher in degraded than in intact habitats<sup>18</sup>, suggesting cases where colonisation by winners may be stronger than local extinction of losers.

These three possible trajectories – where species richness either systematically decreases, remains stable, or increases with human pressure – all assume that individual species responses to human pressure are monotonous. However, this response can take very different shapes<sup>3</sup>, with many species being favoured by an intermediate level of human pressure but negatively affected when human pressure becomes too high (e.g., bird species associated with agricultural habitats that are negatively impacted by agriculture intensification<sup>19</sup>). As a result, we could expect that some winner species will go locally extinct (or at least stop colonising) above a certain human pressure, resulting in a threshold in species richness response to human pressure<sup>20</sup>. The above-mentioned insensitivity of species richness to human pressure would thus not hold when human pressure becomes too high, as suggested by some local studies<sup>21-23</sup>. Similarly, in cases of systematic increases of species richness with human pressure, we could expect species richness to reach a peak at intermediate pressure, before declining. This somewhat echoes the intermediate disturbance hypothesis, which suggests that species coexistence may be maximal at disturbance with intermediate frequency or intensity<sup>24,25</sup>. As species richness is a key index in ecology and conservation, we need to increase and unify our understandings on the exact impact of human pressure on biodiversity.

Here I take advantage of three large-scale occurrence biodiversity datasets to study the effect of human pressure on local assemblages by comparing the richness and composition of assemblages in

sites with different level of human pressure. First, I apply a modelling method that enables discriminating between all the above-mentioned possible trajectories of species richness with increasing human pressure, with subsets from eBird in eight tropical forest biodiversity hotspots<sup>26</sup>, from the North America Breeding Bird Survey<sup>27</sup> and from the PREDICTS database<sup>28</sup> (Fig. 1). Second, I explore how assemblage composition in the eBird data is impacted by human activities and how this explains trends in species richness (Fig. 2). Specifically, I characterise assemblage composition based on its specialisation to forest habitats (measuring richness in species with high/medium dependence, *vs* species with low or no dependence on forest habitats), its endemicity (measuring richness in endemic species, *vs* large-range species), its sensitivity to human activities (measuring richness in high-sensitivity species, *vs* tolerant and anthropophilic species), and its global dynamic (measuring richness in threatened and Near Threatened species, *vs* non-native species). Combining these results on overall species richness and composition, I propose a unifying framework on species richness response to human pressure (Fig. 3).

All analyses consisted in two successive models. First, the "ecological model" controlled for variation in diversity indices that is not due to human pressure, controlling for sampling heterogeneity (e.g., duration of sampling events, number of observers) and ecological heterogeneity (altitude, latitude, and Net Primary Productivity) between samples. Then, the "footprint model" used the residuals from the "ecological model" to measure the specific effect of human footprint on diversity indices (see details in Methods).

# Results

### Human footprint impact on overall species richness

The relationship between human footprint and local overall species richness of eBird checklists showed a significant threshold. It consisted in a first phase where overall species richness slightly increased with human footprint, followed by a second phase of steep decline, when human footprint exceeded 33.8 (Fig. 1A; Table S1). This threshold broadly corresponds to urbanised and highly intensive agricultural habitats (note that the model could not estimate a threshold >33.9 which corresponds to the ninth decile of the response variable; see details in Methods). Compared with a linear model, this threshold model enabled reducing the AIC by 368 (Table S2) and the pattern was supported by General Additive Models with smoothed response curves (i.e., with fewer *a priori* assumptions; Supplementary Figures S4-5).

When hotspots were considered separately, the above-mentioned two-phase relationship emerged very clearly from the hotspots with the highest statistical power (Mesoamerica and Tropical Andes, representing respectively 50% and 27% of sampling events) as well as Indo-Burma and Sundaland (Fig. 1B; Table S1). A threshold was also significant in the Atlantic Forest with a first phase with no significant trend followed by a steep decline. Finally, I fitted linear models in three hotspots, because thresholds were not significant, and found that species richness was increasing with human footprint in the Eastern Afromontane (coef=0.013, P=0.019) and Western Ghats and Sri Lanka (coef=0.009, P<10<sup>-4</sup>), and did not change significantly in Tumbes-Chocó-Magdalena (coef=0.003, P=0.37; Fig. 1B, Table S1).

Species richness in the PREDICTS database did not show a significant threshold. A linear model was thus fitted and presented a significant decline with human footprint (coef=-0.004, P<10<sup>-4</sup>; Fig. 1C, Table S1), and a lower AIC than the threshold model (Table S2). When considering taxa separately, the threshold was significant for none of them (respectively P=0.36, P=0.53, P=0.15 for vertebrates, invertebrates and plants) and species richness decreased linearly for invertebrates (coef=-0.0071, P<10<sup>-4</sup>), tended to decrease for plants (coef=-0.0034, P=0.088) and was stable for vertebrates (coef=-0.0016, P=0.32). I observed the same pattern when looking only at birds (N=1644, threshold P-value=0.76, linear model P-value=0.38) and birds in tropical regions (N=892, threshold P-value=0.19, linear model P-value=0.88). Species richness in BBS data showed a similar pattern as the one observed in eBird Atlantic Forest checklists, with a first phase with no significant trend until a threshold estimated at 25.45 and followed by a steep decline (Fig. 1D; Table S1).



Figure 1: Effect of human footprint on overall species richness for each region and dataset. A: eBird data across the eight tropical forest biodiversity hotspots. B: eBird data modelled independently for each hotspot. C: PREDICTS data across the globe. D: BBS data across the US and Canada. Each curve represents the selected model: threshold model if significant (blue; two p-values are provided: the first one corresponds to the test of the first slope compared to 0, the second corresponds to the test of the difference of the second slope compared to the first slope), a linear model otherwise (purple; a single p-value, corresponding to the slope compared to 0). Detailed statistics in Tables S1-2 and fit in Fig. S3. To facilitate interpretation, the y-axis does not present residuals of 'ecological models' but rather the predicted species richness for a standard sampling event (see details in Methods). Histograms (grey bars) represent the distribution of human footprint values of sampling sites for each region (distribution of species richness values is given in Supplementary Figure 2). P-values: NS>0.05>\*>0.01>\*\*>0.001>\*\*\*.

Results were consistent when using the human footprint maps of 2009<sup>29</sup> and 2013<sup>30</sup>, showing similar patterns but with some variations in the value where thresholds occurred (Supplementary Figures S6-7, Supplementary Tables S3-4).

## Changes in assemblage composition in eBird data

Forest specialists, endemic species, threatened and Near Threatened species and high-sensitivity species are overall negatively affected by human footprint: they are "losers" of the ongoing global increase in human footprint (Fig. 2, Table S1). Conversely, species with low and null dependence to forest habitats, species with large ranges, non-native species and tolerant or anthropophilic species are mainly positively affected by human footprint: they are "winners" (Fig. 2, Table S1). More specifically, in terms of forest-specialisation (Fig. 2A, Table S1), I found that species with high dependence on forest habitats present the steepest decline (linear decline; coef=-0.042, P<10<sup>-10</sup>), followed by species with medium dependence that show a low decline, accelerating when human footprint exceeds 32.3. Conversely, both non-forest species and species with low dependence on forest habitats see their richness increase up to a human footprint of ca. 24 before reaching a plateau. In terms of endemism (Fig. 2B, Table S1), richness in endemic species declines with human footprint, especially above a threshold of 30.7, while richness in species with a large range increases up to a threshold of 33.7 before declining. Richness in threatened and Near Threatened species declines with human footprint, especially when human footprint exceeds 7.5. Conversely, richness in non-native species increases with human footprint, especially when it exceeds 31.1 (Fig. 2C, Table S1). Finally, in terms of sensitivity to human activities (Fig. 2D, Table S1), I found that richness in high-sensitivity species declines linearly with human footprint (coef=-0.049, P<10<sup>-10</sup>). Conversely, richness in tolerant species increases with human footprint, up to a limit of 31.6 and then decreases while richness in anthropophilic species increases continuously, especially when human footprint is below 6.8.

Results were consistent when using the human footprint maps from Venter et al.<sup>29</sup> and Williams et al.<sup>30</sup> (Supplementary Figures S8-9, Supplementary Tables S3-4).



Figure 2: Effect of human footprint on eBird assemblage composition measured by threshold models for 11 indices of assemblage composition. Two p-values are provided when threshold was significant (the first p-value corresponds to the test of the first slope compared with 0, the second p-value corresponds to the test of the second slope compared with the first slope). When threshold was not significant, a linear model was fitted and a single p-value is provided (corresponding to the test of the slope compared with 0). Detailed statistics in Tables S1-2 and fit in Fig. S3. To facilitate interpretation, the y-axis does not present residuals of 'ecological models' but rather the predicted species richness for a standard sampling event (see details in Methods). P-values: NS>0.05>\*>0.01>\*\*\*.

# Discussion

In this study, I used a simple methodology to identify a diversity of trajectories in species richness response to human pressure, that I apply to three large-scale empirical datasets. The method aims at isolating human impact on species richness, by controlling for natural drivers of variation in species richness, but this remains imperfect as this only based on correlations. I find that species richness response to human pressure can present important thresholds, with a first phase where species richness is poorly influenced by human pressure although assemblage composition is deeply transformed, followed by a second phase of steep decline in species richness. More importantly, the diversity of trajectories evidenced in this study enables delineating a general unifying framework (the "Replace then remove framework") which includes five main trajectories that are described and differ by two characteristics: the slope of the first phase and the linearity of species response to human pressure (Fig. 3).

With the eBird data, I find a slight increase in species richness during the first phase (+ 5 species when human footprint increases from 0 to 33.5), typical of the Intermediate peak trajectory (i.e.,

increase in species richness, followed by a decline; Fig. 3). This positive trend hides a strong transformation of assemblage composition due to the replacement of "losers" by "winners", that I highlight in four comparisons (Fig. 2). First, forest habitat specialists (typical from the studied regions; i.e., tropical forests) decline while species that are associated with other habitat types increase with human footprint, consistently with the biotic homogenisation literature<sup>5,6</sup>. Second, species that are endemic to the studied hotspots decline while species with large range size increase with human footprint<sup>6,8,9</sup>. Third and intuitively, species that are threatened or Near Threatened (i.e., those that have been internationally recognised as negatively affected by human activities) decline with human footprint, while settled non-native species increase with human footprint<sup>31</sup>. Finally, richness in high-sensitivity species declines with human footprint, conversely to tolerant and anthropophilic species. This last result might suffer from partial circularity as the species sensitivity index was elaborated with similar data<sup>3</sup>; but circularity may be weakened by the larger scale considered to create the sensitivity index (most data are not used in the present study) and the difference in response variable (individual species abundances were used to create the index vs richness in a certain category of species here). This first phase thus consists in a strong transformation of assemblage composition with a slight positive effect on species richness, highlighting once again that species richness is a poor index to measure human effects on local biodiversity<sup>13,14,17,26</sup>.



Figure 3: Theoretical plot presenting the "Replace then remove framework" developed in this study and its five trajectories. The two axes present the two characteristics to look at to identify which trajectory applies to a given situation; the slope of the first phase and the strict monotony of the trajectory.

Generalising these findings, I suggest that the first phase of the five trajectories in the Replace then remove framework consists in the replacement of losers by winners, and that the overall slope of species richness depends on the relative rates of colonisation and extinction. If colonisation by winners is higher than extinction by losers, the Systematic increase (i.e., consistent increase in species richness with human pressure) or the Intermediate peak trajectory can occur. I found the former pattern in Eastern Afromontane and Western Ghats and Sri Lanka, and the later in the eBird global dataset and four hotspots, echoing the Intermediate Disturbance Hypothesis at a macroecological scale<sup>25</sup>. Conversely, if extinction exceeds colonisation rate in this first phase, a Systematic decline can occur (i.e., monotonous decline in species richness with human pressure; Fig. 3), as I evidenced with the PREDICTS database, and has been reported in previous studies<sup>11,12</sup>. Finally, in some cases, the colonisation of winners perfectly offsets the extinction rate of losers, leading to a flat slope in the first phase in the Plateau trajectory (i.e., flat trend in species richness, followed by a steep decline) and the Species replacement trajectory (i.e., flat trend in species richness across the human footprint gradient; Fig. 3). Such balance is surprisingly common in the literature<sup>13-16</sup>, although methods of these studies have been subject to criticism<sup>32</sup>, and was reported in the present study (with eBird data in Tumbes-Choco-Magdalena and PREDICTS data for vertebrates for the Species replacement trajectory, and in the Atlantic Forest and the BBS analyses for the Plateau trajectory; Fig. 1, Table S1).

In the second phase of the Intermediate Peak trajectory with eBird data, I found that many "winners" are not benefiting anymore from the increase in human pressure above a certain threshold (mainly occurring in built environments and highly intensive agriculture). As a result, richness in "winners" declines or plateaus, leading to a drop in local overall species richness (Figs. 1-2). This was evidenced for species with low or null dependence on forest habitats, with large range, or with tolerance to human activities. Only richness in anthropophilic and non-native species increases continuously with human footprint, consistently with the fact that they are largely associated with urban ecosystems<sup>4,33</sup>. Several previous case studies evidenced similar threshold patterns in species richness response to habitat loss or habitat fragmentation<sup>21–23,34</sup>. Different and complex ecological processes can explain the existence of such thresholds, for instance if species are resilient to habitat loss does not lead to immediate extinction but creates an extinction debt<sup>20</sup>. By investigating assemblage composition changes across the human pressure gradient (Fig. 2), the present study highlights the importance of understanding the diversity of individual species response to pressure in order to better understand the emergence of species-richness patterns.

Thresholds are however not systematic and their existence depends on landscape characteristics (e.g., habitat variance or habitat fragmentation) and species characteristics (e.g., dispersal ability, sensitivity to habitat edges, local pool of anthropophilic species that can colonise)<sup>20</sup>. In the Replace then remove framework, I describe two trajectories without thresholds (Species replacement and Systematic increase; Fig. 3), while Systematic decrease can occur with or without a significant threshold. In addition, thresholds are likely undetected in some cases because they are not tested for<sup>20</sup> or due to a low statistical ability to detect them. Indeed, the value of the threshold may vary depending on studied regions, habitats and taxa (e.g., 33.7 for eBird data in tropical forest hotspots and 25.5 with BBS data in North-America) and can be difficult to detect if too high. It can also happen that the threshold is not particularly high, but sampling effort does not include sites with high human footprint (i.e., if only the first phase of the trajectory is correctly sampled). When using temporal datasets, not only do sampling sites need to be located in areas with high pressure, but

they need to cover severe degradation of habitat in the study period (e.g., conversion from seminatural to urban landscapes) to detect such effect on species richness, which may currently be possible only with a handful of datasets.

In all trajectories, assemblages may be strongly transformed, even at low levels of human pressure. Thus, regardless of the impact of human pressure on species richness, it is important to note that the composition and functioning of assemblages may be affected, possibly resulting in biotic homogenisation (e.g., with the extinction of endemic, specialist, threatened species, replaced by generalist, large-ranged, non-native species). This may result in increased functional homogeneity<sup>5</sup>, with the possible eviction of species with key role in the assemblage such as predators, or specific pollinators. These effects may worsen in the second phase, where assemblages lose species. In addition to these compositional changes, the decline of species richness in the second phase may translate in a decline in overall abundance of assemblages, or defaunation<sup>35–37</sup>, suggesting that species resources are too limited in these degraded habitats and that carrying capacity is thus declining.

# Conclusion

In this study, I propose a framework unifying five different trajectories of species richness response to human pressure – the Replace then remove framework – that describes five trajectories articulated around two characteristics of trajectories. This framework highlights the importance of considering assemblage composition in studies aiming at measuring the impact of human activities or conservation actions, as species richness can hide strong assemblage transformations<sup>14,38</sup>. It also highlights the importance of considering non-linear and non-monotonous assumptions when investigating the effects of human activities on biodiversity in order to detect existing thresholds. Overall, these results provide strong evidence that human pressure deeply transforms assemblages, probably resulting in biotic homogenisation, but that we still have a lot more to lose if human footprint keeps increasing globally.

# Methods eBird data

#### Bird counts

eBird is an online platform gathering >700 million bird observations across the globe<sup>39,40</sup>. Observations are made opportunistically but are gathered in checklists and are accompanied with information on sampling effort, allowing for the transformation of data into a standard dataset and the study of assemblage composition across large spatial scales<sup>41,42</sup>. In a previous study, Cazalis et al.<sup>26</sup> created a standard subset of the eBird database, keeping only complete checklists (i.e., for which observers certified having reported all species detected) collected between 2005 and 2018 (I here only kept the checklists made between 2010 and 2018 and later control for year in analyses), arising from sampling events that lasted between 0.5-10 hours, either from "stationary points" or "travelling counts" when distance travelled was <5km, and reported by observers with sufficient experience. Finally, they transformed the eBird data to fit the bird taxonomy used by BirdLife International in order to be able to use BirdLife International species characteristics.

The study area used in Cazalis et al.<sup>26</sup> and here, included eight tropical forest biodiversity hotspots<sup>43</sup> that correspond to areas with exceptional biodiversity and high level of habitat loss. They are thus highly relevant places to measure the impact of human activities. The study area was restricted to the part of biodiversity hotspots that were included in the "Tropical and subtropical moist broadleaf forests" biome<sup>44</sup>, to maximise sites comparability. It included the eight biodiversity hotspots that presented sufficient eBird data after filtration, with 4 in the Americas (Atlantic Forest, Tropical Andes, Tumbes-Chocó-Magdalena, Mesoamerica), 1 in Africa (Eastern Afromontane), and 3 in Asia (Western Ghats and Sri Lanka, Indo-Burma, Sundaland). Of the 66,777 sampling events available from the above-mentioned study, I only kept the 62,945 that were done in 2010 or after, and for which I was able to calculate every landscape covariate (see Fig. 1 for the number of sampling events per hotspot).

#### Sampling covariates

Because data collection in eBird is not standardised, I calculated several covariates of sampling effort and quality following Cazalis et al.<sup>26</sup>, later used to control for sampling heterogeneity in all analyses. I used the duration of sampling (in minutes) and the number of observers, as the probability to detect species increases with time and with number of observers. I also used an observer calibration index calculated in Cazalis et al.<sup>26</sup> and developed from Kelling et al.<sup>45</sup> and Johnston et al.<sup>46</sup>, that incorporates heterogeneity in observers' ability to detect species because of their difference in expertise, equipment and behaviours. Finally, I used the day of the year of observation to control for temporal trends in bird diversity.

#### Landscape covariates

To quantify human pressure, I used the global human footprint index, built from the combination of spatial information on human infrastructures, agricultural land use and population density. Two different datasets of human footprint have been released in the last years with similar methodologies: one mapping the human footprint in 1993 and 2009<sup>29</sup> and the other one mapping the human footprint in 2000, 2005, 2010, and 2013<sup>30</sup>. There are important variations in values between

these two datasets. Indeed, the variation when comparing two different years of a single dataset (the difference between the 2000 and 2013 values from Williams et al.<sup>30</sup> in sampled sites was  $0.7 \pm 1.7$ ) was much lower than the variation when comparing two similar years from the different datasets (the difference in Williams et al.<sup>30</sup> value in 2010 and Venter et al.<sup>29</sup> in 2009 was  $2.7 \pm 4.0$ ). To ensure my results are robust to the measure of human footprint, I therefore used a combined measure in the analyses, and test for the sensitivity of my results to the dataset used (Supplementary Figures S6-9, Supplementary Tables S3-4). As eBird data used in this study were collected between 2010 and 2018, I used the most recent version of both datasets, and calculated the average between the median value of human footprint pixels intersecting with a 1-km buffer around the sampling site in 2009 and in 2013. The remaining temporal mismatch may add some noise in the results and reduce significance (e.g., if human footprint of study sites has greatly increased between human footprint and eBird data collection).

I calculated a value of altitude and Net Primary Productivity for each sampling site. Altitude data were obtained from the GLOBE Digital Elevation Model<sup>47</sup>, which has a 0.008 degree resolution (ca. 930m at latitude 0). I calculated the altitude of each sampling site as the median of the values intersecting a 1-km buffer around the sampling site. Net Primary Productivity was calculated from a raster averaging monthly productivity maps from Jan.2014 to Nov.2016<sup>48</sup> and extracting the median of the values intersecting a 1-km buffer around the sampling site.

## Assemblage indices

For each of the 65,465 checklists I calculated twelve assemblage indices. Overall species richness was the total number of bird species detected in the checklist. I used BirdLife International<sup>49</sup> 4levels classification of species dependence on forest habitats to characterise the habitat specialisation of assemblages, measuring for each checklist: the richness in species with high dependence on forest habitats, medium-dependence, low-dependence and "Non-Forest species". I used BirdLife International and HBW<sup>50</sup> distribution maps to characterise species endemicity. I measured for each checklist the number of species that were endemic to the hotspot (i.e., >90% of their range in the hotspot; see Cazalis et al.<sup>26</sup>), and the number of large-range species (i.e., that had a global distribution >1,000,000 km<sup>2</sup>). I used BirdLife International<sup>49</sup> Red List status to measure the number of threatened and Near Threatened species of each checklist and BirdLife International and HBW<sup>50</sup> to measure the number of non-native species of each checklist, considering as non-native to a given hotspot every species that only has "Introduced" distribution within the hotspot. Finally, I used an index of bird sensitivity to human footprint developed by Cazalis et al.<sup>3</sup> to measure the sensitivity of assemblages to human (only for hotspots from the Americas). For each checklist from the Americas, I measured the number of species characterised as "High-sensitivity" (i.e., species whose sensitivity index is in the highest quartile of the 4,424 species breeding in the Americas), the number of species characterised as "Anthropophilic" (i.e., species whose sensitivity index is in the 5% lowest values; broadly corresponding to species whose abundance increases continuously with human pressure), and the number of species that are "Tolerant" (i.e., species whose sensitivity index is in the lowest quartile, excluding "Anthropophilic" species).

## Breeding Bird Survey data

I investigated the link between local species richness and human footprint using the North America Breeding Bird Survey (BBS). This scheme consists in yearly sampling by volunteer birders around routes of ca. 25 miles with 50 count points recording every bird detected<sup>51</sup>. Specifically, I used the data processed in a recent study<sup>27</sup> consisting in 3,016 BBS routes, only considering the section including the first 5-stops (to ensure linking local human pressure conditions with local bird assemblages) and combining 5 years of observations randomly selected between 2007 and 2016 (to ensure assemblages are representative of the local diversity). These routes are well distributed in the US and the Southern part of Canada and cover different habitat types (forest, shrubland, cropland, grassland, mosaic, bare area, urban, and others)<sup>27</sup>. I calculated the human footprint (using the average human footprint between the 2009 map from Venter et al.<sup>29</sup> and the 2013 map from Williams et al.<sup>30</sup>, to match with observation dates) within a 500m buffer around the first section of BBS routes, as in Cazalis et al.<sup>27</sup>, and used the value of Net Primary Productivity from the same study.

#### **PREDICTS** data

Finally, I used the PREDICTS – Projecting Responses of Ecological Diversity In Changing Terrestrial Systems – database<sup>52</sup>, from which Newbold et al.<sup>11</sup> found a systematic decline in species richness with human pressure. It gathers data from scientists worldwide in order to produce a global database of terrestrial species' responses to human pressure. It now includes over 3.6 million biodiversity records from > 32,000 sites, covering > 50,000 species of diverse taxa. Specifically, I used the public release of the database developed in Gray et al.<sup>28</sup> consisting in 6,531 sampling sites across most realms (33% in Palearctic, 17% in Neotropical, 15% in Afrotropical, 13% in Indomalayan, 12% in Australasian, and 9% in Nearctic), taxa (40% vertebrates, 37% invertebrates, 23% plants), and land-use (38% of minimal use, 39% of light use, 17% of intense use, and 6% undetermined). After removing the 1.3% of studies that started before 2000, I obtained 6,448 sites, all starting sampling between 2000 and 2012. I used the altitude value from the original dataset and calculated the human footprint (as the average between 2009 map from Venter et al.<sup>29</sup> and the 2005 map from Williams et al.<sup>30</sup>) and Net Primary Productivity (using the same map as with eBird data) from a 1-km buffer around the sites coordinates.

#### Statistical analyses

To measure the effect of human footprint on each diversity index, I first created a model (that I call "ecological model") for each diversity index, aiming at controlling for ecological effects and sampling heterogeneity. I then created a second model (called "footprint model"), using the residuals of the ecological model as response variable, to study the effects of human footprint. The results from the footprint models cannot be taken as causal evidence as the model is based on correlation, but I expect the ecological model to control for most variation in species richness that is not due to human activities.

#### Controlling for ecological factors

For the three datasets, I used a common model structure, including as predictors: altitude to account for elevational physiological constraints on species and also past patterns of speciation (with a quadratic effect, allowing non-monotonous response of species richness to altitude) and Net Primary Productivity to account for climatic factors (with a quadratic effect). In addition, I controlled for latitude (assuming a quadratic effect as species richness is likely to peak around latitude 0) for PREDICTS and BBS data, but not for eBird as this information is mostly contained in the control for hotspots (which cover relatively narrow ranges of latitude). In addition, I controlled for several sampling factors, specific to each dataset.

For each eBird diversity index I created an "ecological model" running a GAM<sup>53</sup> with a similar structure to Cazalis et al.<sup>26</sup> and assuming negative binomial distributions. I controlled for 7 continuous variables and 1 factor variable: altitude, Net Primary Productivity, year (linear assumption to account for general temporal trends in response variables, e.g., temporal decline of sensitive species), day of the year (smoothed), duration (smoothed with limited curve complexity), observer calibration index (smoothed with limited curve complexity), number of observers (smoothed with limited curve complexity), and hotspot name (to account for differences in bird diversity indices across regions). The ecological model had the following structure:

 $gam(Index \sim scale(altitude) + scale(altitude)^2 + scale(npp) + scale(npp)^2 + year + s(Day) + s(Duration, k=4) + s(Observer_Calibration, k=4) + s(N_observers, k=4) + Hotspot_name)$ 

For the BBS database, I created an "ecological model" with a GAM model, assuming a negative binomial distribution of the species richness, with the following structure: gam(Richness ~ scale(altitude) + scale(altitude)^2 + scale(npp) + scale(npp)^2 + scale(latitude) + scale(latitude)^2)

For the PREDICTS database, following Gray et al.<sup>28</sup>, I created an "ecological model" using a mixed Generalised Linear Model (using the function *glmer* from the *lme4* package<sup>54</sup>), accounting for heterogeneity between studies. It assumed a Poisson distribution and had the following structure, including random effects on the study (SS), study block (SSB), and study sites (SSBS):

glmer(Richness ~ scale(altitude) + scale(altitude)^2 + scale(npp) + scale(npp)^2 + scale(latitude) + scale(latitude)^2 + (1|SS|) + (1|SSB) + (1|SSBS)). To investigate the consistency of the results across taxa, I also ran this model independently for studies focusing vertebrates, invertebrates, plants, birds, and tropical birds.

#### Measuring human footprint impact on overall species richness

For each dataset, I built a "footprint model" to link the residuals of the ecological model on overall species richness with human footprint (see distributions in Fig. 1), able to identify the different possible trajectories suggested in the introduction. I used the *chngptm* function from the R package *chngpt*<sup>55</sup> using a segmented threshold, fitting a function with two join segments. The model tests for the absolute trend of the first segment (i.e., is the trend different from zero), estimates a threshold value (found between the first and ninth deciles of the explanatory variable), and tests for the difference of the second trend (i.e., providing the difference in slope with the first segment and testing whether the change in slope is significant). A significant threshold enables identifying cases of "Intermediate peak trajectory", where richness peaks at intermediate human footprint (first trend significant, followed by a steep decline). When the threshold was not significant, I instead used a classic linear model, using the *lm* function from the R package *stats*<sup>56</sup>, even if the threshold model

presented a lower AIC (Table S2). Such linear models enable identifying cases of Systematic decline (trend significantly negative), Systematic increase (trend significantly positive) or Species replacement (trend not significant; Fig. 3).

In addition, I tested for the consistency between hotspots in the eBird data by applying the same modelling approach to each hotspot independently.

The plots presented in Figs. 1-2 result from the use of the sum predictions (R function *predict*) of the "ecological model" and the "footprint model", from a dataset with every quantitative variable fixed to their median value in the dataset. The hotspot name variable in eBird models was fixed to "Atlantic Forest" and the three variables used as random effects with the PREDICTS database fixed to their first value in the dataset.

## Measuring changes in assemblage composition in eBird data

Using eBird data, I tested the impact of human footprint on assemblage composition by modelling the relationship between human footprint and the residuals of ecological models for the eleven assemblage composition indices. I first fitted a threshold model and, when the second segment was not significant (i.e., there is no significant threshold), I replaced it by a simple linear model.

# Acknowledgments:

I thank Elie Gaget and Ana S.L. Rodrigues for their insightful and constructive comments on previous versions of the manuscript. I am also grateful to the numerous observers who contributed records to eBird, and to the eBird team at Cornell Lab of Ornithology for creating and managing this database, as well as the thousands of U.S. and Canadian participants who annually perform and coordinate the Breeding Bird Survey, and the contributors and managers of the PREDICTS database. Time and effort were supported by sDiv, the Synthesis Centre of iDiv (DFG FZT 118, 202548816).

**Data and materials availability:** Data and code used in this study is available at <u>https://doi.org/10.5281/zenodo.6345865</u>.

# References

- IPBES. Global assessment report on biodiversity and ecosystem services of the Intergovernmental Science- Policy Platform on Biodiversity and Ecosystem Services. E. S. Brondizio, J. Settele, S. Díaz, and H. T. Ngo (editors). (2019).
- Dornelas, M. et al. A balance of winners and losers in the Anthropocene. Ecology Letters 22, 847–854 (2019).
- 3. Cazalis, V. et al. Mismatch between bird species sensitivity and the protection of intact habitats across the Americas. Ecology Letters **24**, 2394–2405 (2021).
- Guetté, A., Gaüzère, P., Devictor, V., Jiguet, F. & Godet, L. Measuring the synanthropy of species and communities to monitor the effects of urbanization on biodiversity. Ecological Indicators 79, 139–154 (2017).
- Clavel, J., Julliard, R. & Devictor, V. Worldwide decline of specialist species: toward a global functional homogenization? Frontiers in Ecology and the Environment 9, 222–228 (2011).
- Rutt, C. L., Jirinec, V., Cohn-Haft, M., Laurance, W. F. & Stouffer, P. C. Avian ecological succession in the Amazon: A long-term case study following experimental deforestation. Ecology and Evolution 9, 13850–13861 (2019).
- McKinney, M. L. & Lockwood, J. L. Biotic homogenization: a few winners replacing many losers in the next mass extinction. Trends in Ecology & Evolution 14, 450–453 (1999).
- Newbold, T. et al. Widespread winners and narrow-ranged losers: Land use homogenizes biodiversity in local assemblages worldwide. PLOS Biology 16, e2006841 (2018).
- Finderup Nielsen, T., Sand-Jensen, K., Dornelas, M. & Bruun, H. H. More is less: net gain in species richness, but biotic homogenization over 140 years. Ecology Letters 22, 1650–1657 (2019).
- Staude, I. R. et al. Replacements of small- by large-ranged species scale up to diversity loss in Europe's temperate forest biome. Nat Ecol Evol 4, 802–808 (2020).
- 11. Newbold, T. et al. Global effects of land use on local terrestrial biodiversity. Nature 520, 45-50

(2015).

- Murphy, G. E. P. & Romanuk, T. N. A meta-analysis of declines in local species richness from human disturbances. Ecology and Evolution 4, 91–103 (2014).
- Dornelas, M. et al. Assemblage time series reveal biodiversity change but not systematic loss. Science 344, 296–299 (2014).
- 14. Hillebrand, H. et al. Biodiversity change is uncoupled from species richness trends: Consequences for conservation and monitoring. Journal of Applied Ecology 55, 169–184 (2018).
- Sax, D. F., Gaines, S. D. & Brown, J. H. Species Invasions Exceed Extinctions on Islands Worldwide: A Comparative Study of Plants and Birds. The American Naturalist 160, 766–783 (2002).
- Vellend, M. et al. Global meta-analysis reveals no net change in local-scale plant biodiversity over time. PNAS 110, 19456–19459 (2013).
- 17. Supp, S. R. & Ernest, S. K. M. Species-level and community-level responses to disturbance: a cross-community analysis. Ecology **95**, 1717–1723 (2014).
- Hiley, J. R., Bradbury, R. B. & Thomas, C. D. Impacts of habitat change and protected areas on alpha and beta diversity of Mexican birds. Diversity Distrib. 22, 1245–1254 (2016).
- 19. Reif, J. & Vermouzek, Z. Collapse of farmland bird populations in an Eastern European country following its EU accession. Conservation Letters **12**, e12585 (2019).
- 20. Swift, T. L. & Hannon, S. J. Critical thresholds associated with habitat loss: a review of the concepts, evidence, and applications. Biological Reviews **85**, 35–53 (2010).
- Muylaert, R. L., Stevens, R. D. & Ribeiro, M. C. Threshold effect of habitat loss on bat richness in cerrado-forest landscapes. Ecological Applications 26, 1854–1867 (2016).
- 22. Maron, M. et al. Spurious thresholds in the relationship between species richness and vegetation cover. Global Ecology and Biogeography **21**, 682–692 (2012).
- 23. Simmonds, J. S., Rensburg, B. J. van, Tulloch, A. I. T. & Maron, M. Landscape-specific

thresholds in the relationship between species richness and natural land cover. Journal of Applied Ecology **56**, 1019–1029 (2019).

- Connell, J. H. Diversity in Tropical Rain Forests and Coral Reefs. Science 199, 1302–1310 (1978).
- Roxburgh, S. H., Shea, K. & Wilson, J. B. The intermediate disturbance hypothesis: patch dynamics and mechanisms of species coexistence. Ecology 85, 359–371 (2004).
- Cazalis, V. et al. Effectiveness of protected areas in conserving tropical forest birds. Nature Communications 11, 4461 (2020).
- 27. Cazalis, V., Belghali, S. & Rodrigues, A. S. L. Using a large-scale biodiversity monitoring dataset to test the effectiveness of protected areas at conserving North-American breeding birds. Peer Community Journal 1, e3 (2021).
- Gray, C. L. et al. Local biodiversity is higher inside than outside terrestrial protected areas worldwide. Nature Communications 7, 12306 (2016).
- Venter, O. et al. Global terrestrial Human Footprint maps for 1993 and 2009. Scientific Data 3, 160067 (2016).
- Williams, B. A. et al. Change in Terrestrial Human Footprint Drives Continued Loss of Intact Ecosystems. One Earth 3, 371–382 (2020).
- Ellis, E. C., Antill, E. C. & Kreft, H. All Is Not Loss: Plant Biodiversity in the Anthropocene.
  PLOS ONE 7, e30535 (2012).
- 32. Cardinale, B. J., Gonzalez, A., Allington, G. R. H. & Loreau, M. Is local biodiversity declining or not? A summary of the debate over analysis of species richness time trends. Biological Conservation 219, 175–183 (2018).
- McKinney, M. L. Urbanization as a major cause of biotic homogenization. Biological Conservation 127, 247–260 (2006).
- Andrén, H. Habitat Fragmentation, the Random Sample Hypothesis and Critical Thresholds. Oikos 84, 306–308 (1999).

- 35. Dirzo, R. et al. Defaunation in the Anthropocene. Science 345, 401–406 (2014).
- WWF. Living Planet Report 2018. Aiming higher. Grooten, M. and Almond, R.E.A. (Eds).
  WWF, Gland, Switzerland. (2018).
- 37. Ceballos, G., Ehrlich, P. R. & Dirzo, R. Biological annihilation via the ongoing sixth mass extinction signaled by vertebrate population losses and declines. PNAS 201704949 (2017) doi:10.1073/pnas.1704949114.
- McGill, B. J., Dornelas, M., Gotelli, N. J. & Magurran, A. E. Fifteen forms of biodiversity trend in the Anthropocene. Trends in Ecology & Evolution 30, 104–113 (2015).
- Sullivan, B. L. et al. eBird: A citizen-based bird observation network in the biological sciences. Biological Conservation 142, 2282–2292 (2009).
- 40. Sullivan, B. L. et al. The eBird enterprise: An integrated approach to development and application of citizen science. Biological Conservation **169**, 31–40 (2014).
- 41. Strimas-Mackey, M. et al. Best Practices for Using eBird Data. Version 1.0. https://cornelllabofornithology.github.io/ebird-best-practices/. Cornell Lab of Ornithology, Ithaca, New York. https://doi.org/10.5281/zenodo.3620739. (2020).
- 42. Johnston, A. et al. Analytical guidelines to increase the value of community science data: An example using eBird data to estimate species distributions. Diversity and Distributions n/a, (2021).
- 43. Mittermeier, R. A. et al. Hotspot revisited. 390 (2004).
- 44. Olson, D. M. et al. Terrestrial Ecoregions of the World: A New Map of Life on EarthA new global map of terrestrial ecoregions provides an innovative tool for conserving biodiversity. BioScience 51, 933–938 (2001).
- 45. Kelling, S. et al. Can Observation Skills of Citizen Scientists Be Estimated Using Species Accumulation Curves? PLOS ONE **10**, e0139600 (2015).
- 46. Johnston, A., Fink, D., Hochachka, W. M. & Kelling, S. Estimates of observer expertise improve species distributions from citizen science data. Methods in Ecology and Evolution **9**,

88-97 (2018).

- 47. National Geophysical Data Center. Global Land One-kilometer Base Elevation (GLOBE), version 1. <a href="https://www.ngdc.noaa.gov/mgg/topo/gltiles.html">https://www.ngdc.noaa.gov/mgg/topo/gltiles.html</a>. (1999) doi:10.7289/V52R3PMS.
- 48. NASA Earth Observatory Team. Net Primary Productivity (TERRA / MODIS) maps. (2020).
- 49. BirdLife International. IUCN Red List for birds. Version 2017.1. downloaded from <a href="http://www.birdlife.org">http://www.birdlife.org</a>>. (2017).
- BirdLife International and HBW. Bird species distribution maps of the world. Version 7.0.
  Available at <a href="http://datazone.birdlife.org/species/requestdis">http://datazone.birdlife.org/species/requestdis</a>. (2017).
- 51. Pardieck, K. L., Ziolkowski, D. J., Lutmerding, M., Campbell, K. & Hudson, M.-A. R. North American Breeding Bird Survey Dataset 1966 - 2016, version 2016.0. U.S. Geological Survey, Patuxent Wildlife Research Center. <a href="https://www.pwrc.usgs.gov/bbs/RawData/">https://www.pwrc.usgs.gov/bbs/RawData/</a>; doi:10.5066/F7W0944J. (2017) doi:10.5066/F7W0944J.
- 52. Hudson, L. N. et al. The database of the PREDICTS (Projecting Responses of Ecological Diversity In Changing Terrestrial Systems) project. Ecol Evol 7, 145–188 (2017).
- 53. Wood, S. N. Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models: Estimation of Semiparametric Generalized Linear Models. Journal of the Royal Statistical Society: Series B (Statistical Methodology) 73, 3–36 (2011).
- Bates, D., Mächler, M., Bolker, B. & Walker, S. Fitting Linear Mixed-Effects Models Using Ime4. J. Stat. Soft. 67, (2015).
- 55. Fong, Y., Huang, Y., Gilbert, P. & Permar, S. chngpt: threshold regression model estimation and inference. BMC Bioinformatics DOI:10.1186/s12859-017-1863-x, (2017).
- R Core Team. R: A Language and Environment for Statistical Computing. (R Foundation for Statistical Computing, 2020).